Author's response to reviews

Title: The Manchester Color Wheel: development of a novel way of identifying color choice and its validation in healthy, anxious and depressed individuals

Authors:

Helen R Carruthers (helen.carruthers@uhsm.nhs.uk)
Julie Morris (julie.morris@manchester.ac.uk)
Nicholas Tarrier (nicholas.tarrier@manchester.ac.uk)
Peter J Whorwell (peter.whorwell@manchester.ac.uk)

Version: 6 Date: 26 January 2010

Author's response to reviews:

Dear Sir/Madam,

We have revised the paper according to your latest request. All the methods and all the results have been brought together under 'Methods' and 'Results'. We have added a conclusion at the end of the discussion and the tables have now created using a the tables tool.

We hope that all this meets with your approval.

Kind regards,

Peter Whorwell